Author's response to reviews

Title: Oedema of the metatarsal heads II-IV and forefoot pain as unusual manifestation of Lyme disease

Authors:

Stefan Endres (s.endres@elisabeth-klinik.de)
Markus Quante (quante@med.uni-marburg.de)

Version: 4 Date: 20 June 2007

Author's response to reviews:

Dear ladies and gentlemen,

thank you for the quick accept of our case report. Dimitrios Filippou, one of the reviewer wanted to update the references. But also if there are a lot of newer publications dealing with the lyme disease they weren't necessary for the present case report. Nevertheless we have added some newer publications.

The other reviewer, Richard Weiner, has accepted the case report in its present form.

At least the manuscript has been overworked according the guidelines of the Journal of medical case reports.

Best regards

Dr. Stefan Endres